**Author’s response to reviews**

**Title:** Proenkephalin A 119-159 (penKid) – a novel biomarker for acute kidney injury in sepsis: An observational study.

**Authors:**

Mari Rosenqvist (mari.rosenqvist@med.lu.se)
Kevin Bronton (kevin.bronton@med.lu.se)
Oliver Hartmann (hartmann@sphingotec.de)
Andreas Bergmann (bergmann@sphingotec.de)
Joachim Struck (struck@sphingotec.de)
Olle Melander (olle.melander@med.lu.se)

**Version:** 2 **Date:** 22 Oct 2019

**Author’s response to reviews:**

In the section 'Ethics Approval and Consent to Participate', please state whether the informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

Author response: Renamed heading. Elaborated on consent procedure.

In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Author response: Elaborated on the role and influence on funding bodies in the design, execution of and interpretation of the study.

Please rename the Ethics Approval title to 'Ethics approval and consent to participate', Introduction to 'Background' &amp; MATERIALS AND METHODS to 'Methods'.

Author response: Renamed and reformatted headings accordingly.

Thank you for providing a Letter to Editor. As this document is no longer required at this stage of the publication process, please remove it from your submission’s supplementary files.

Author response: Letter to Editor removed from supplementary files.